Author's response to reviews

Title: New Jersey Center for Tourette Syndrome Sharing Repository: methods and sample description

Authors:

Gary A Heiman (heiman@biology.rutgers.edu)
Robert A King (robert.king@yale.edu)
Jay A Tischfield (jay@Biology.Rutgers.Edu)

Version: 6 Date: 26 November 2008

Author's response to reviews:

November 26, 2008

Editorial Production Team
BioMed Central

Reference MS #: 1425713254196704 - New Jersey Center for Tourette Syndrome Sharing Repository: methods and sample description

Editorial Production Team,

I have made two changes to the manuscript:

1) In the first paragraph of the background, I changed a comma to a semicolon (after ADHD).

2) I have re-written the second sentence of the section entitled “Sample description” (page 6). It now reads, “In addition, we are currently assembling a research team whose goal is to submit a multi-site collaboration grant to substantially increase the sample size.”

Other than that, there are no other changes.

Sincerely,

Gary A. Heiman, Ph.D.
Assistant Research Professor
Rutgers University